Author's response to reviews

Title: Shwachman-Diamond Syndrome: A complex case demonstrating the potential for misdiagnosis as Asphyxiating Thoracic Dystrophy (Jeune Syndrome)

Authors:

Steven J Keogh (sjackeogh@me.com)  
Shane McKee (shane.mckee@belfasttrust.hscni.net)  
Sarah F Smithson (Sarah.Smithson@uhbristol.nhs.uk)  
David Grier (davidgrier@btinternet.com)  
Colin G Steward (colin.steward@bristol.ac.uk)

Version: 3 Date: 31 January 2012

Author's response to reviews: see over
Dear Editor,

Re: MS: 1296551209583895 - Shwachman-Diamond Syndrome: A complex case demonstrating the potential for misdiagnosis as Asphyxiating Thoracic Dystrophy (Jeune Syndrome)

Thank you to the editorial team and reviewer for the suggestions for improvements to our paper. We have now revised our manuscript accordingly, this involved putting arrows on Figures 2 through 6 to show up the essential features and we amended the legend accordingly.

We hope that these changes meet with your approval and are grateful for your further consideration of our manuscript for publication.

Kind Regards,

Dr Steven Keogh